Author’s response to reviews

Title: Retinal vasculitis with Chronic Recurrent Multifocal Osteomyelitis: a case report and review of the literature

Authors:

Victoria Shanmugam (vshanmugam@mfa.gwu.edu)
Marc Phillpotts (mphillpotts@mfa.gwu.edu)
Timothy Brady (tim_brady@gwmail.gwu.edu)
Monica Dalal (mdalal@mfa.gwu.edu)
Shawn Haji-Momenian (shajimomenian@mfa.gwu.edu)
Esma Akin (eakin@mfa.gwu.edu)
Kavita Nataranjan (knataranjan@mfa.gwu.edu)
Sean McNish (mcnish23@gmail.com)
Donald Karcher (dkarcher@mfa.gwu.edu)

Version: 3 Date: 11 Mar 2019

Author’s response to reviews:

Dear Dr. Gaffo,

RE: Retinal vasculitis with Chronic Recurrent Multifocal Osteomyelitis: a case report and review of the literature (BRHM-D-18-00087)

Thank you for reviewing the attached manuscript for possible publication in the BMC Rheumatology. We have responded to reviewer comments below.

1. Rename the heading Case Summary to “Case presentation”

This change has been made

2. Include three to ten keywords representing the main content of the article below the abstract and above the main text
The keywords have been moved as requested.

3. Confirm consent

This has been further outlined as requested.

4. Funding section, please clarify the role of the funding body in the design of the study, collection, analysis and interpretation of the data and in writing the manuscript.

This has been clarified as requested but please note this study was not funded.

5. Please represent authors names using their full initials not their full name in the authors contributions section. If there are duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

This has been done.

6. Individual contributions of all authors to the manuscript should be specified in the Author Contributions section. Currently TB is not present.

This has been added.

7. CARE checklist is not required please remove.

This has been removed as requested.

Thank you for reconsidering this revised manuscript for publication.

Yours sincerely,

Victoria K. Shanmugam, MD, MRCP

Associate Professor of Medicine